**Author's response to reviews**

**Title:** Infraglenoidal scapular notching in reverse total shoulder replacement: a prospective series of 60 cases and systematic review of the literature

**Authors:**

Patrick Sadoghi (patricksadoghi@gmx.at)
Andreas Leithner (andreasleithner@medunigraz.at)
Patrick Vavken (patrick.vavken@childrens.harvard.edu)
Andreas Hölzer (andreas.hoelzer@med.uni-muenchen.de)
Josef Hochreiter (josef.hochreiter@bhs.at)
Georg Weber (georg.weber@bhs.at)
Matthias F Pietschmann (matthias.pietschmann@med.uni-muenchen.de)
Peter E Müller (peter.mueller@med.uni-muenchen.de)

**Version:** 4  **Date:** 17 May 2011

**Author's response to reviews:**

Dear Editorial Board!

Please find our answers according to your requests as followed:

**Formatting changes to be made:**

-------------------------------

**Major revisions (we require the author to make these changes)**

Figures: All figures must be cited; figures not cited will be excluded from the final manuscript.

--> All figures were cited in the manuscript. Figure 1 had already been cited in the patients and methods section. Figures 2 to 7 are now further cited in the results section:

Radiological data in terms of infraglenoidal scapular notching which had been classified according to Nerot et al. [9] are presented in table 2 and figures 2 to 7.

**Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)**

--> we have made some of these changes

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

--> we have removed all highlighting and tracking from the manuscript.
Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

--> we would kindly ask you to change this as requested

References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

--> we would kindly ask you to add this as requested

Tables- please remove the visible vertical lines from your tables

--> we would kindly ask you to remove this as requested

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

--> we have checked the typography of the manuscript. No further corrections were made.

With kind regards,
Patrick Sadoghi, MD